Author’s response to reviews

Title: The influence of active coping and perceived stress on health disparities in a multi-ethnic low income sample

Authors:

Jennifer M Watson (jwatson@dental.ufl.edu)
Henrietta L Logan (Hlogan@dental.ufl.edu)
Scott L Tomar (Stomar@dental.ufl.edu)

Version: 4 Date: 11 January 2008

Author’s response to reviews: see over
Dear Editorial Team:

Thank you for the letter outlining necessary formatting changes prior to publication of the manuscript entitled: “The Influence of Active Coping and Perceived Stress on Health Disparities in a Multi-Ethnic Low Income Sample” ms #1138791797157296, to BioMed Public Health. We have made all the requested formatting changes and these changes are outlined below.

Authors’ Affiliations have been changed to the proper format: department, institute, city, and country.

All underlining has been removed and replaced with either bold/italics/indentation.

References have been cited in the text using consecutive numbers in square brackets any references with superscripts were corrected and all references were placed before the full stop.

A Conclusions header was added above the final paragraph of the discussion.

Authors Contributions was changed to Authors' contributions

Table cells were reformatted to include borders and clear cells and fields, text alignment was corrected to ensure it is clear and consistent.

Every table is referenced in order in the manuscript and a reference to Table one was added to the Participants section.

All reference were cited in the text before the full stop.

We greatly appreciate the opportunity to publish this important work in BioMed Public Health. Thank you for the careful review.

Sincerely,

Jennifer Watson, PhD